Author’s response to reviews

Title: The promotion of oral health within the Healthy School context in England: a qualitative research study

Authors:

   Emma Stokes (estokes@liv.ac.uk)
   Cynthia M Pine (c.pine@salford.ac.uk)
   Rebecca V Harris (r.v.harris@liv.ac.uk)

Version: 4 Date: 9 January 2009

Author’s response to reviews:

The resubmitted paper: ‘The promotion of oral health within the Healthy School context in England: a qualitative research study’ has been amended as advised by the paper’s reviewers and by editors of BMC Oral Health. The precise amendments are listed below:

The spelling mistake on page 6 under paragraph ‘study design’ has been corrected.

The role of the Local Healthy School Coordinator has been explained for the benefit of readers not from UK.

The reporting of the results has been made more precise by the provision of more numerical data.

The meaning of all abbreviations has been explained.

A statement has been inserted to indicate that all the authors read and approved the final manuscript.

Vertical lines have been removed from Tables 2.

Section headings have been modified to use sentence case

Underlining has been removed.

Reference 5 has been corrected.